Author's response to reviews

Title: Development of a practical guide for the early recognition of malignant melanoma of the foot and nail unit

Authors:

Ivan R Bristow (ib@soton.ac.uk)
David AR de Berker (david.deberker@ubht.nhs.uk)

Version: 6 Date: 27 September 2010

Author's response to reviews:

One comma was misaligned by one character space and has been adjusted.